Check for updates

OPEN ACCESS

EDITED BY Teresa Strisciuglio, University of Naples Federico II, Italy

REVIEWED BY

Carlo Lavalle, Sapienza University of Rome, Italy Vincenzo Mirco La Fazia, Texas Cardiac Arrhythmia Institute, United States

*CORRESPONDENCE Xian Wang wxian202212@163.com

RECEIVED 26 February 2024 ACCEPTED 07 May 2024 PUBLISHED 16 May 2024

CITATION

Ma L, Wang B, Fan J, Zhou H, Li J, Li W, Zheng X and Wang X (2024) Cardiocerebrovascular benefits of early rhythm control in patients with atrial fibrillation detected after stroke: a systematic review and meta-analysis.

Front. Cardiovasc. Med. 11:1391534. doi: 10.3389/fcvm.2024.1391534

COPYRIGHT

© 2024 Ma, Wang, Fan, Zhou, Li, Li, Zheng and Wang. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

Cardiocerebrovascular benefits of early rhythm control in patients with atrial fibrillation detected after stroke: a systematic review and meta-analysis

Liang Ma¹, Baofu Wang^{1,2}, Jiasai Fan¹, Hufang Zhou¹, Jingen Li^{1,3}, Weisheng Li², Xiangying Zheng¹ and Xian Wang^{1*}

¹Department of Cardiology, Dongzhimen Hospital, Beijing University of Chinese Medicine, Beijing, China, ²Department of Integrative Medicine Cardiology, China-Japan Friendship Hospital, Beijing, China, ³Department of Cardiovascular Medicine, Mayo Clinic, Rochester, MN, United States

Objective: This study aimed to evaluate the impact of early rhythm control (ERC) on the occurrence of cardiocerebrovascular events in patients diagnosed with atrial fibrillation detected after stroke (AFDAS).

Methods: A systematic search was conducted across nine databases from inception to October 15, 2023 to identify clinical trials comparing ERC with usual care interventions in AFDAS patients. The primary outcome assessed was recurrent stroke, with secondary outcomes including all-cause mortality, adverse events related to arrhythmias, and dementia.

Results: Analysis of five studies, consisting of two randomized clinical trials (RCTs) involving 490 patients and three cohort studies involving 95,019 patients, revealed a reduced rate of recurrent stroke [odds ratio (OR) = 0.30, 95% confidence interval (CI) 0.11–0.80, P = 0.016 in RCTs; OR = 0.64, 95% CI 0.61–0.68, P < 0.00001 in cohort studies] and all-cause mortality (hazards ratio = 0.94, 95% CI 0.90–0.98, P = 0.005 in cohort studies) in the ERC group compared to the usual care group. In addition, ERC was associated with superior outcomes in terms of dementia.

Conclusions: Patients with AFDAS who underwent ERC treatment exhibited a decreased risk of cardiocerebrovascular events compared to those receiving usual care. These results support the potential benefits of implementing an ERC strategy for this specific patient population.

Systematic Review Registration: https://www.crd.york.ac.uk/PROSPERO/, ldentifier [CRD42023465994].

KEYWORDS

early rhythm control, atrial fibrillation, stroke, cardiocerebrovascular events, recurrent stroke, all-cause mortality

1 Introduction

Atrial fibrillation (AF) is characterized by uncoordinated electrical conduction in the atria, leading to uncoordinated atrial systole and diastole. The estimated global prevalence is over 37 million (1). AF can result in a range of cardiocerebrovascular diseases, including stroke, heart failure, dementia, and ultimately death. Notably, AF significantly increases

the risk of stroke by three to five times, highlighting its importance as a key risk factor (2). Conversely, stroke can also trigger the detection of AF, with approximately 1.5 million stroke survivors worldwide later being diagnosed with AF (3). Bernstein et al. discovered that the rate of detecting AF in patients with a history of stroke is 21.7% within a 3-year period (4). Nevertheless, there is a noticeable absence of standardized treatment guidelines for patients with AFDAS (5).

Current therapeutic approaches for AFDAS involve anticoagulant medications to reduce the risk of cardiovascular and cerebrovascular complications. To manage newly diagnosed AF, it is recommended that rate control treatment be prioritized in cases where symptoms manifest. If this approach does not provide any relief, antiarrhythmic drugs (AADs) and ablation procedures must be considered (5). However, recent research suggests that oral anticoagulation therapy in AFDAS may not significantly reduce the risk of stroke or systemic arterial embolism (6, 7). Sposato et al. found that the burden of AF in AFDAS was more severe compared to AF alone (3). Due to poststroke neurogenic mechanisms, stroke-induced heart injury, autonomic dysfunction, and inflammatory responses, individuals with AFDAS often have a higher burden of AF, increasing the risk of recurrent stroke and exacerbating cardiovascular and cerebrovascular diseases (8, 9). Therefore, in addition to anticoagulation therapy, AFDAS patients must explore proactive antiarrhythmic treatment strategies to improve their prognosis.

The 2020 EAST-AFNET 4 trial revealed that early rhythm control (ERC) for AF patients leads to better long-term outcomes compared to usual care (UC), in which patients were primarily treated with rate control without rhythm control (10). There were no significant differences in adverse effects between the two approaches (10). However, the clinical benefit of rhythm control therapy for AFDAS patients is still uncertain. Due to the associated risks of stroke, thromboembolism, and atrioesophageal fistula following catheter ablation (11, 12), there is a tendency in clinical practice to avoid prescribing rhythm control therapy for AFDAS considering the real clinical benefit and cost-effectiveness ratio (13).

Therefore, this study aims to investigate the impact of strengthened rhythm control in the early stages of AFDAS on cardiovascular and cerebrovascular outcomes, providing valuable evidence on the effectiveness of ERC for AFDAS.

2 Methods

2.1 Registration

This review adhered to the reporting guidelines for systematic reviews and was registered with PROSPERO, an international prospective register of systematic reviews (CRD42023465994) (14).

2.2 Search strategy

To search for relevant articles, we conducted search in nine databases, namely, PubMed, Embase, Cochrane Library, Web of

Science, Clinical Key, ClinicalTrials.gov, CNKI, Wan Fang, and VIP databases, up to October 15, 2023 for relevant clinical trials investigating rhythm control in stroke-complicating AF. The search strategy, detailed in Supplementary Material, was based on combinations of keywords such as "stroke," "atrial fibrillation," "rhythm control," and their synonyms.

2.3 Inclusion and exclusion criteria

The inclusion criteria for the study literature included the following: (1) subjects with a history of stroke and newly diagnosed with AF; (2) the experimental group receiving early AF diagnosis (within 1 year) received antiarrhythmic therapy, whereas the control group received usual treatment; (3) studies reporting adverse events related to cardiovascular and cerebrovascular diseases, including recurrent stroke, all-cause mortality, adverse events related to arrhythmias, and dementia; and (4) unrestricted by age, gender, race, etc.

The exclusion criteria were as follows: (1) non-clinical and irrelevant studies; (2) studies with unpublished or missing data from registered clinical trial results; (3) in case of studies with repetitive or similar data sources, retaining the most comprehensive and relevant report or literature; and (4) clinical trial protocols, reviews, meta-analyses, case reports, and conference abstracts.

2.4 Literature screening and data extraction

Duplicate literature was excluded, and two researchers independently conducted a preliminary review based on titles and abstracts, followed by a full-text review. Relevant information on study design and outcomes was extracted for final inclusion in this study. Two researchers independently performed data extraction, including first author, publication date, sample size, gender, followup period, interventions, and outcome indicators. Disagreements were resolved by discussion or, if necessary, with the assistance of a third researcher. The two researchers then independently conducted a full-text assessment to confirm the key information of the included studies. Finally, the extracted information was crosschecked by both researchers, and any discrepancies were resolved through consultation with a third researcher.

2.5 Quality assessment

Two quality-control reviewers assessed the quality of the literature using the Cochrane Handbook's quality assessment criteria for randomized clinical trials (RCTs) and the Newcastle–Ottawa Scale (NOS) for non-randomized controlled trials. Both quality-control officers independently assessed the quality and the risk of bias of each study. In case of a disagreement, a third reviewer performed a reassessment.

2.6 Outcomes and definitions

The primary outcome of this study was recurrent stroke. The secondary outcomes included all-cause mortality, adverse events

related to arrhythmias, and dementia. Recurrent stroke was defined as incident stroke during follow-up. Incident stroke was determined by the primary diagnosis during admission using ICD-10-CM codes.

2.7 Statistical methods

Data analysis was performed using Stata14.0. Considering the limited number of included studies, a fixed-effects model was employed (15). Odds ratios (ORs) were used to analyze binary variables, while hazards ratios (HRs) were used for all-cause mortality. P < 0.05 was considered statistically significant.

3 Results

3.1 Study selection

Using a computer system to search the nine databases mentioned above, a comprehensive compilation of 21,102 pertinent pieces of studies on ERC for AFDAS has been amassed as of October 15, 2023. A total of 16,061 duplications were excluded from consideration. A subsequent scrutiny of titles and abstracts led to the exclusion of 5,008 articles that failed to meet the stipulated criteria. Finally, after an exhaustive review of the entire article, five studies were included for the purpose of metaanalysis and systematic review (Figure 1).

3.2 Basic information of studies

Five eligible studies (16–20) were published in English, involving 95,509 participants, with 22,928 in the ERC group and 7,581 in the UC group. This included two RCTs (18, 19) with 490 patients and three cohort studies (16, 17, 20) with 95,019 patients. Four studies reported recurrent stroke and mortality rates (17–20), while only one study reported dementia (16). The longest follow-up time for each study was taken as the outcome information collection point for research.

For the implementation of treatment strategies, the ERC group received either rhythm control surgery or drug treatment within 1 year of AF diagnosis. Noteworthy distinctions emerged in the approach to the UC group across studies. In their investigation, Lee et al. (16, 17) refrained from administering rhythm control therapy to patients during the observation period, whereas Sagris et al. (20)



Outcomes		 Recurrent stroke Mortality 3. Adverse events related to arrhythmia 	 Recurrent stroke Mortality 3. Adverse events related to arrhythmia 	 Recurrent stroke Mortality 	 Recurrent stroke Mortality 	 All dementia Alzheimer dementia Vascular dementia 	; NOAC, non-vitamin K
Max follow-up (years)			œ	10	æ	8	i, usual care
tion	U	UC	UC	лс	UC	UC	ion; UC
Interven		ERC	ERC	ERC	ERC	ERC	eter ablat
latelets	U	34 (35.8)	29 (27.1)	33 (40.2)	NR	NR	s and cath
Antipl		61 (34.3)	17 (15.5)	48 (35.6)	NR	NR	ling AADs
agulants C/VKA)	U	89 (93.7)	92 (86.0)	37 (45.1)	41,135 (100)	NR	:rol, incluc
Anticoa (NOA		166 (93.3)	106 (96.4)	33 (40.2)	12,284 (100)	NR	thm cont
)S2-VASc ore	U	4.3 (1.6)	5.0 (1.5)	3.4 (1.5)	5.5 (1.6)	NR	early rhy
CHA2-E		4.4 (1.5)	5.0 (1.5)	3.4 (1.5)	5.5 (1.6)	NR	trial; ERC,
iistent n (%)	U	47 (49.5)	30 (28.0)	NR	NR	NR	d clinical
Pers AF,		84 (47.2)	29 (26.4)	NR	NR	NR	ndomizeo
xysmal n (%)	U	48 (50.5)	36 (33.6)	87 (100)	NR	NR	l; RCT, ra
Paro AF,		94 (52.8)	49 (44.5)	143 (100)	NR	NR	C, contro
pisode 1 (%)	U	NR	41 (38.3)	NR	NR	NR	reatment;
First e AF, i		NR	32 (29.1)	NR	NR	NR	iation; T, t
mean /ears)	U	70.1 (7.8)	72 (7.8)	75.3 (8.1)	72 (10.9)	(11)	ndard dev
Age, (SD,)		68.9 (8.9)	71 (7.4)	74 (8.9)	72.2 (9.6)	20	e; SD, stal
(%)	U	64.2	47	56.3	53	~	ombin
Male		60.7	41	51.7	52.9	ŝ	annot c gonist.
aple ze	U	95	107	87	41,135	31,157	rted or câ iin K anta
San si:		178	110	143	12,284	10,213	not repo 'KA, vitarr
Study design		RCT	RCT	Cohort study	Cohort study	Cohort study	ካ (%); NR, ჰgulant; V
Include studies (years)		Park et al. (19)	Jensen et al. (18)	Sagris et al. (20)	Lee et al. (17)	Lee et al. (16)	Values are , oral anticos

04

studies.	
included	
of the	
ristics o	
characte	
Basic o	
TABLE 1	

Study	Random sequence generation	Allocation concealment	Blinding of participants and personnel	Blinding of outcome assessment	Free of infrequent missing outcome data	Free of selective reporting	Other bias
Park et al. (19)	Probably yes ^a	Definitely no Open-label	Definitely no Open-label	Definitely yes ^b	Definitely yes ^c	Probably yes ^d	Probably yes ^d
Jensen et al. (18)	Definitely yes ^e	Definitely no Open-label	Definitely no Open-label	Definitely yes Blinded outcome assessment	Definitely yes ^d	Probably yes ^d	Probably yes ^d

TABLE 2 Risk of bias assessment of included randomized controlled trials.

^aMethod for generating a randomization sequence was not clearly reported. We judged that randomization sequence generation was likely not achieved given that it used a one-to-one basis according to instructions.

^bThe blind method was not used, but the outcome indicators were not affected.

^cNo more than patients assigned to early rhythm control had no information about the history of stroke and were excluded from this subgroup analysis, but it is unlikely to affect the outcomes.

^dGenerally balanced baseline characteristics across groups.

^eRandomization was performed by an independent statistician using a central randomization list.

and Jensen et al. (18) initially applied rate control therapy to the UC group. If adequate rate control failed to restore the sinus rhythm (SR) or provide symptomatic relief, antiarrhythmic control therapy was introduced. In the work of Park et al. (19), usual care involved incorporating antiarrhythmic therapy 2 months after enrollment, depending on the patient's specific condition. The foundational data across all investigations depicted a semblance of comparability, with patients having an average age range of 68–75 years. Table 1 summarizes the basic characteristics reported in each study.

3.3 Risk of bias

In two RCTs (18, 19), the study by Jensen et al. (18) randomized participants using a central randomization list and blinded the outcomes. Park et al. (19) did not provide a detailed description of the random allocation method, and one subject was excluded from both the experimental and control groups. Due to the use of rhythm control methods, such as radiofrequency ablation in both studies, blinding of subjects was not possible, but the impact on outcome indicator detection was not significant. Tables 2 and 3 provide additional information.

Three cohort studies (16, 17, 20) were prospective. For both the studies of Lee et al. (17) and Sagris et al. (20), relevant medical records were extracted. In the study of Sagris et al. (20), the control group consisted of patients with AF that could spontaneously revert to sinus rhythm, while the experimental group consisted of patients who could revert to sinus rhythm after AF treatment, consequently affecting the intergroup comparability. At the same time, 15.2% of the patients dropped out, which may affect the outcome indicators. Therefore, no score was given for the relevant items. Table 3 displays more detailed items in each domain. The NOS scores of the three studies were 9 (17), 6 (20), and 6 (16), respectively.

3.4 Primary outcome

3.4.1 Recurrent stroke

Four studies (17–20), comprising two RCTs (18, 19) and two cohort studies (17, 20), assessed the incidence of recurrent stroke.

The meta-analysis found that the recurrence stroke was lower in the ERC group than in the UC group (OR = 0.64; 95% CI 0.60– 0.68; P < 0.00001; Figure 2). In the RCTs, the meta-analysis showed a statistically significant reduction in the occurrence of recurrent strokes in the ERC group than in the UC group after treatment (OR = 0.30, 95% CI 0.11–0.80, Z = 2.40, P = 0.02). In the cohort study, the meta-analysis showed a statistically significant lower incidence of recurrent stroke events in the ERC group than in the UC group (OR = 0.64, 95% CI 0.61–0.68, Z = 15.05, P < 0.00001).

3.5 Secondary outcome

3.5.1 All-cause mortality rate

Three studies (17, 18, 20), including one RCT (18) and two cohort studies (17, 20), analyzed all-cause mortality. In the RCT (18), all-cause mortality within 8 years of treatment was lower in the ERC group [11% (10%)] than in the UC group [21% (20%)]. In cohort studies, the HR for all-cause mortality was statistically significant, that is, HR = 0.94, 95% CI 0.90–0.98, P = 0.005, as shown in Figure 3. This illustrates a significantly reduced risk of all-cause mortality in the ERC group.

3.5.2 Adverse events related to arrhythmia

Regarding the safety aspects of rhythm control treatment, one study (18) reported adverse events associated with rhythm control therapy. The results showed a higher incidence of adverse events in patients receiving ERC than in to those receiving UC [3 (3%) vs. 2 (2%)]. These events included drug-induced bradycardia, syncope attributed to rhythm control therapy, and implantation of a pacemaker or other cardiac devices in the ERC group. In the UC group, there were two events related to hospitalization for AF and drug toxicity of AF-related drug therapy.

Moreover, another study (19) reported arrhythmia-related events. The results showed that there was no statistically significant difference in the incidence of ERC compared to UC [5 (2.8%) vs. 1 (1.1%); P = 0.372]. These adverse arrhythmia events included pacemaker implantations (permanent or temporary) in patients with potential sinus disease (four cases in total) and a single case of syncope.

		01	Selection		Comparability		Outcome	
Study	Representativeness of the cohort	Selection of the non-exposed cohort	Ascertainment of exposure	Demonstration that outcome of interest was not present at the start of the study	Comparability of cohorts on the basis of design or analysis time to follow-up	Assessment of outcome	Follow-up long enough for outcomes to occur	Adequacy of follow-up cohorts
Lee et al. (17)	-	1	1	1	2	-	1	-1
Sagris et al. (20)		0 _a	1	1	1 ^b	-	1	0,
Lee et al. (16)	-	1	0q	1	2	0q	1	0q
The patien	ts in the control group all have	e AF that spontaneously	converts to sinus rhythr	Э.				

The baseline characteristic of the coronary history was different (P < 0.05).

Forty-five patients (15.2%) were lost to the scheduled follow-up (from 6 to 75 months).

¹Not mentioned

One cohort study (16) reported the incidence of dementia in AFDAS patients. The result showed that ERC treatment was superior to UC treatment in reducing the incidence of dementia, Alzheimer's disease, and vascular dementia (HR: 0.27 vs. 0.33; 0.05 vs. 0.06; 0.21 vs. 0.26).

10 3389/fcvm 2024 1391534

4 Discussion

3.5.3 Dementia

This systematic review and meta-analysis revealed that ERC may decrease the risk of recurrent stroke in patients with AFDAS. In addition, ERC was also associated with a lower risk of all-cause mortality and dementia. This study represents the initial systematic review and meta-analysis investigating the effectiveness and safety of ERC in patients with AFDAS.

This study included five studies with a total of 95,509 participants, comparing ERC with UC for patients with AFDAS in relation to cardiocerebrovascular events. The findings indicate that ERC can reduce the risk of recurrent stroke and the incidence of all-cause mortality and alleviate the occurrence of dementia. There was no significant difference in arrhythmiarelated adverse event rates between the two groups.

Recent studies have shown that AFDAS can be categorized into neurogenic atrial fibrillation (NAF) and cardiogenic atrial fibrillation (CAF), each with distinct pathological mechanisms that contribute to the occurrence of atrial fibrillation (8, 21). Sposato et al. (7) found that, after excluding the influence of confounding factors such as oral anticoagulants, when comparing patients with AFDAS, stroke with SR, and AF known before the stroke (KAF), there was no statistically significant difference in the 1-year ischemic stroke recurrence rates between AFDAS and SR. Interestingly, the highest incidence of recurrent stroke was observed in KAF. This indicated that the timing of AF onset in stroke patients may be influenced by different pathophysiological factors, including cardiogenic and neurogenic origins.

NAF may result from damage to the dorsal anterior insular cortex, which regulates the parasympathetic nerve after ischemic stroke. This damage can cause an imbalance in the heart and the autonomic nervous system, leading to an upregulated sympathetic nerve activity and a subsequent myocardial injury (22, 23). Compared to SR, no difference was found in stroke incidence; therefore, for NAF, the antiarrhythmic therapy can be appropriately relaxed. However, it is more worthy to monitor changes in heart function as the incidence of acute heart failure or worsening heart failure is 3.6 times compared to KAF (6). ByIn contrast, CAF in AFDAS may stem from cardiac pathology, a well-known high-risk factor for stroke. CAF could potentially precede stroke onset but remain undetected due to the technical limitations of ECG detection. The duration of AF has been identified as a significant factor influencing the risk of stroke (24). AFDAS patients, especially those undetected with AF before the stroke, may have longer AF duration, leading to a higher incidence of stroke and adverse events related to cardiovascular and cerebrovascular events. Therefore, effective antiarrhythmic

TABLE 3 Risk of bias assessment of included cohort studies



control therapy, such as the use of ERC, is necessary for this population.

ERC not only alleviates the symptoms of AF but also reduces the atrial fibrillation burden and the hospitalization time (10). Its positive impact extends to cardiac structural and functional recuperation, promoting hemodynamic stability, preventing thrombosis, reducing recurrent stroke rates and adverse events, and enhancing cardiac output and cerebral blood perfusion. These benefits are crucial for protecting the cognitive function (25). Although there is currently a lack of controlled studies comparing NAF and CAF, we believe that ERC for AFDAS may provide greater clinical benefits for patients, regardless of



whether they have NAF or CAF, given the potential stroke prevention and cardiac function preservation.

The impact of stroke duration on decision-making for ERC treatment in AFDAS remains unexplored. Regrettably, our study lacked sufficient data for subgroup analysis, limiting our ability to make definitive conclusions. However, Song et al. found that performing radiofrequency ablation surgery on AF patients within 3 months of stroke resulted in similar postoperative complication rates as those without a stroke history. Moreover, there was a reduced risk of recurrent stroke after 12 months of follow-up compared to usual care (26). This evidence suggests that ERC could be a safe and effective treatment for AF patients who recently suffered a stroke within 3 months.

Regarding specific rhythm control treatment choices, no study is currently analyzing the difference between AADs and radiofrequency ablation in patients with AFDAS. Although ablation tools and procedures have advanced, radiofrequency ablation may still carry a risk of subclinical or clinical thromboembolism with an estimated stroke risk of 0.5%–1% during the procedure (27). AADs also have drawbacks, including adverse reactions like arrhythmia and QT interval prolongation, and often require personalized dose adjustments and dynamic monitoring. As direct comparison is lacking, clinicians typically make treatment decisions based on individual patient's characteristics, such as age, heart function, stroke history, and other complications. In some cases, a combination of both AADs and radiofrequency ablation may be necessary to achieve optimal rhythm control.

ERC can decrease cerebrovascular incidence in AFDAS patients and benefit other frail patients affected by AF. Currently, catheter ablation is the first choice in ERC treatment. La Fazia et al. (28) found that human immunodeficiency virus (HIV) patients with AF have more serious inflammation and oxidative stress caused by HIV infection, which can contribute to atrial remodeling and AF burden (28). For HIV + AD patients, undergoing catheter ablation earlier can significantly reduce the incidence of atrial arrhythmias. Sohns et al. (29) found that for end-stage HF with AF, CA was associated with a lower rate of all-cause mortality.

4.1 Comparison with previous studies

This study is the first systematic review and meta-analysis to examine the treatment benefits of ERC in patients with AFDAS. The investigation used a meticulous approach and conducted multiple searches across various online platforms to ensure a comprehensive and exhaustive search strategy. The quality assessment was reasonable, and the included literature exhibited a high level of homogeneity.

4.2 Limitation

Numerous constraints necessitate careful consideration when interpreting our findings. First, the scope of included studies was limited, preventing us from conducting meta-analyses for AFrelated subtypes and other adverse events. While we performed a comprehensive literature search for all eligible trials, this limitation may introduce some bias into our study. Furthermore, potential biases may have arisen in the timing of rhythm control treatments within the UC group. It is also possible that some patients with ischemic stroke and KAF had undiagnosed atrial fibrillation in their medical history. Moreover, the complexities of ECG presented technical challenges, making it difficult to determine whether arrhythmias in stroke-complicated AF patients are of cardiac or neurogenic origin.

5 Conclusion

The use of ERC in patients with AFDAS has shown superior outcomes compared to usual care in terms of reducing the risk of cardiocerebrovascular events. However, considering the limited number of studies included in our analysis, the results of this study need to be further validated.

Data availability statement

The original contributions presented in the study are included in the article/Supplementary Material; further inquiries can be directed to the corresponding author.

Author contributions

LM: Formal analysis, Software, Writing – original draft, Writing – review & editing. BW: Investigation, Resources, Software, Visualization, Writing – review & editing. JF: Conceptualization, Project administration, Validation, Visualization, Writing – review & editing. HZ: Conceptualization, Project administration, Supervision, Writing – review & editing. JL: Project administration, Resources, Validation, Writing – review & editing. WL: Project administration, Validation, Visualization, Writing – review & editing. XZ: Methodology, Project administration, Resources, Supervision, Writing – review & editing. XW: Data curation, Methodology, Visualization, Writing – review & editing, Writing – original draft.

Funding

The authors declare financial support was received for the research, authorship, and/or publication of this article.

This study was supported by the High-level Key Discipline– Integrated Clinical Discipline of Traditional Chinese and Western Medicine (No. zyyzdxk-2023259).

Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated

References

1. Lippi G, Sanchis-Gomar F, Cervellin G. Global epidemiology of atrial fibrillation: an increasing epidemic and public health challenge. *Int J Stroke*. (2021) 16(2):217–21. doi: 10.1177/1747493019897870

2. Escudero-Martínez I, Morales-Caba L, Segura T. Atrial fibrillation and stroke: a review and new insights. *Trends Cardiovasc Med.* (2023) 33(1):23–9. doi: 10.1016/j. tcm.2021.12.001

3. Sposato LA, Field TS, Schnabel RB, Wachter R, Andrade JG, Hill MD. Towards a new classification of atrial fibrillation detected after a stroke or a transient ischaemic attack. *Lancet Neurol.* (2023) 20(1):110–22. doi: 10.1016/S1474-4422(23)00326-5

4. Bernstein RA, Kamel H, Granger CB, Piccini JP, Katz JM, Sethi PP, et al. Atrial fibrillation in patients with stroke attributed to large- or small-vessel disease: 3-year results from the STROKE AF randomized clinical trial. *JAMA Neurol.* (2023) 80 (12):1277–83. doi: 10.1001/jamaneurol.2023.3931

5. Hindricks G, Potpara T, Dagres N, Arbelo E, Bax JJ, Blomström-Lundqvist C, et al. 2020 ESC guidelines for the diagnosis and management of atrial fibrillation developed in collaboration with the European Association for Cardio-Thoracic Surgery (EACTS): the task force for the diagnosis and management of atrial fibrillation of the European Society of Cardiology (ESC) developed with the special contribution of the European Heart Rhythm Association (EHRA) of the ESC. *Eur Heart J.* (2021) 42(5):373–498. doi: 10. 1093/eurhearti/ehaa612

6. Sun Y, Miller MM, Yaghi S, Henninger N. Association of atrial fibrillation detected after stroke with cardiac dysfunction and features of neurogenic cardiac injury. *J Stroke Cerebrovasc Dis.* (2022) 31(6):106445. doi: 10.1016/j. jstrokecerebrovasdis.2022.106445

7. Sposato LA, Cerasuolo JO, Cipriano LE, Fang J, Fridman S, Paquet M, et al. Atrial fibrillation detected after stroke is related to a low risk of ischemic stroke recurrence. *Neurology.* (2018) 90(11):e924–31. doi: 10.1212/WNL.00000000005126

 Sposato LA, Chaturvedi S, Hsieh CY, Morillo CA, Kamel H. Atrial fibrillation detected after stroke and transient ischemic attack: a novel clinical concept challenging current views. *Stroke.* (2022) 53(3):e94–103. doi: 10.1161/STROKEAHA.121.034777

9. Nishizaki K, Itoh T, Kimura M, Tsushima Y, Shoji Y, Kinjo T, et al. Safety and efficacy of contemporary catheter ablation for atrial fibrillation patients with a history of cardioembolic stroke in the era of direct oral anticoagulants. *J Cardiol.* (2017) 70(1):86–91. doi: 10.1016/j.jjcc.2016.10.001

10. Kirchhof P, Camm AJ, Goette A, Brandes A, Eckardt L, Elvan A, et al. Early rhythm-control therapy in patients with atrial fibrillation. *N Engl J Med.* (2020) 383 (14):1305–16. doi: 10.1056/NEJM0a2019422

11. Di Biase L, Burkhardt JD, Santangeli P, Mohanty P, Sanchez JE, Horton R, et al. Periprocedural stroke and bleeding complications in patients undergoing catheter ablation of atrial fibrillation with different anticoagulation management: results from the role of coumadin in preventing thromboembolism in atrial fibrillation (AF) patients undergoing catheter ablation (COMPARE) randomized trial. *Circulation.* (2014) 129(25):2638–44. doi: 10.1161/CIRCULATIONAHA.113.006426

 Kim YG, Shim J, Boo KY, Kim DY, Lee KN, Choi JI, et al. Slit-based irrigation catheters can reduce procedure-related ischemic stroke in atrial fibrillation patients undergoing radiofrequency catheter ablation. *PLoS One.* (2020) 15(10):e0239339. doi: 10.1371/journal.pone.0239339

13. Calvert P, Gupta D, Lip GYH. Early atrial fibrillation rhythm control after stroke. *Lancet Neurol.* (2023) 22(1):3–4. doi: 10.1016/S1474-4422(22)00470-7

14. Moher D, Liberati A, Tetzlaff J, Altman DG. Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. *Br Med J.* (2009) 339:b2535. doi: 10.1136/bmj.b2535

organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Supplementary material

The Supplementary Material for this article can be found online at: https://www.frontiersin.org/articles/10.3389/fcvm.2024. 1391534/full#supplementary-material

15. Borenstein M, Hedges LV, Higgins JP, Rothstein HR. A basic introduction to fixed-effect and random-effects models for meta-analysis. *Res Synth Methods*. (2010) 1(2):97-111. doi: 10.1002/jrsm.12

16. Lee SR, Choi EK, Lee SW, Han KD, Oh S, Lip GYH. Association between early rhythm control and the risk of dementia in patients with atrial fibrillation and prior history of stroke: a nationwide population-based study. *Eur Heart J.* (2022) 43:556–556. doi: 10.1093/eurheartj/ehab472

17. Lee SR, Choi EK, Lee SW, Han KD, Oh S, Lip GYH. Early rhythm control in patients with incident atrial fibrillation who had a prior stroke. *JACC Clin Electrophysiol.* (2023) 9(7 Pt 2):1121–33. doi: 10.1016/j.jacep.2022.11.021

18. Jensen M, Suling A, Metzner A, Schnabel RB, Borof K, Goette A, et al. Early rhythm-control therapy for atrial fibrillation in patients with a history of stroke: a subgroup analysis of the EAST-AFNET 4 trial. *Lancet Neurol.* (2023) 22(1):45–54. doi: 10.1016/S1474-4422(22)00436-7

19. Park J, Shim J, Lee JM, Park JK, Heo J, Chang Y, et al. Risks and benefits of early rhythm control in patients with acute strokes and atrial fibrillation: a multicenter, prospective, randomized study (the RAFAS trial). *J Am Heart Assoc.* (2022) 11(3): e023391. doi: 10.1161/jaha.121.023391

20. Sagris D, Korompoki E, Ntaios G, Tzeis S, Manios E, Kanakakis J, et al. Sinus rhythm restoration and improved outcomes in patients with acute ischemic stroke and in-hospital paroxysmal atrial fibrillation. *Eur Stroke J.* (2022) 7(4):421–30. doi: 10.1177/23969873221109405

21. Cerasuolo JO, Cipriano LE, Sposato LA. The complexity of atrial fibrillation newly diagnosed after ischemic stroke and transient ischemic attack: advances and uncertainties. *Curr Opin Neurol.* (2017) 30(1):28–37. doi: 10.1097/WCO. 000000000000010

22. Krause T, Werner K, Fiebach JB, Villringer K, Piper SK, Haeusler KG, et al. Stroke in right dorsal anterior insular cortex is related to myocardial injury. *Ann Neurol.* (2017) 81(4):502–11. doi: 10.1002/ana.24906

23. Seiffge DJ, Tagawa M. Insights into atrial fibrillation newly diagnosed after stroke: can the brain rule the heart? *Neurology*. (2018) 90(11):493-4. doi: 10.1212/WNL.000000000005113

24. Van Gelder IC, Healey JS, Crijns H, Wang J, Hohnloser SH, Gold MR, et al. Duration of device-detected subclinical atrial fibrillation and occurrence of stroke in ASSERT. *Eur Heart J.* (2017) 38(17):1339–44. doi: 10.1093/eurheartj/ehx042

25. Li M, Jiang C, Lai Y, Wang Y, Zhao M, Li S, et al. Genetic evidence for causal association between atrial fibrillation and dementia: a Mendelian randomization study. *J Am Heart Assoc.* (2023) 12(16):e029623. doi: 10.1161/jaha.123.029623

26. Song ZL, Wu SH, Zhang DL, Jiang WF, Qin M, Liu X. Clinical safety and efficacy of ablation for atrial fibrillation patients with a history of stroke. *Front Cardiovasc Med.* (2021) 8:630090. doi: 10.3389/fcvm.2021.630090

27. Haeusler KG, Kirchhof P, Endres M. Left atrial catheter ablation and ischemic stroke. *Stroke.* (2012) 43(1):265–70. doi: 10.1161/STROKEAHA.111.627067

28. La Fazia VM, Pierucci N, Mohanty S, Gianni C, Della Rocca DG, Compagnucci P, et al. Catheter ablation approach and outcome in HIV+ patients with recurrent atrial fibrillation. *J Cardiovasc Electrophysiol.* (2023) 34(12):2527–34. doi: 10.1111/ jce.16076

29. Sohns C, Fox H, Marrouche NF, Crijns H, Costard-Jaeckle A, Bergau L, et al. Catheter ablation in end-stage heart failure with atrial fibrillation. *N Engl J Med.* (2023) 389(15):1380–9. doi: 10.1056/NEJMoa2306037